Author’s response to reviews

Title: Noninvasive mechanical ventilation with average volume assured pressure support (AVAPS) in patients with chronic obstructive pulmonary disease and hypercapnic encephalopathy

Authors:

Killen H Briones Claudett (kylenbrio@yahoo.com)
Mónica H Briones Claudett (mhobrincla@yahoo.es)
Miguel A Chung Sang Wong (mchungmd@hotmail.com)
Alberto Nuques Martinez (albertonuques@yahoo.com)
Ricardo Soto Espinoza (ricardosotoespinoza@hotmail.com)
Mayra Montalvo (jmonta2@gmail.com)
Michelle Grunauer Andrade (grunauermichelle@hotmail.com)
Antonio Esquinas Rodriguez (antmesquinas@ono.com)
Gumersindo Gonzalez Diaz (gumersindoj.gonzalez@carm.es)

Version: 2 Date: 9 September 2012

Author’s response to reviews:

Cover Letter
Submission date: 09/09/2012

Richard Albert
Editors in Chief

BMC PULMONARY MEDICINE

We have made the following changes

1.- Placement of provisional registration number test International Committee of Medical Journal Editors (ICMJE), because this is still being broadcast

Placed at the end of the summary

Current Controlled Trials application ref is CCT-NAPN-22581.
Kavita Mistry: Database Editor: Current Controlled Trials
236 Gray’s Inn Road. London, WC1X 8HB. United Kingdom
T: +44 (0)20 3192 2160

We’ll send the ICMJE when we are assigned

2.- The contribution of every author according to the sent format: KBC contributed to the study design; data acquisition, analysis, and interpretation; drafting and revision of the manuscript; and preparation of the final version of the
manuscript submitted. MBC contributed to the study design; analysis, and interpretation; and preparation of the final version of the manuscript submitted. MCHS contributed to the study design; data acquisition, analysis and revision of the manuscript. ANM contributed to the study design; data acquisition, analysis, and revision of the manuscript. RSE contributed to the study design; data acquisition, analysis, and revision of the manuscript. MM drafting and revision of the manuscript preparation of the final version of the manuscript submitted. MGA drafting and revision of the manuscript; preparation of the final version of the manuscript submitted. AER drafting and revision, preparation of the final version of the manuscript submitted. All authors assisted with writing the manuscript, and read and approved the final manuscript.

3.- Competing interests.- The authors declare that we have no conflicts of interest

4.- Financial disclosure.- All the Authors have no relevant financial interest in this manuscript.

Sincerely,

Correspondence and reprints:
*Killen Harold Briones Claudett, MD
Pulmonology Department Chief, Military Hospital. Guayaquil, Ecuador
Department of Respiratory Medicine, Panamericana Clinic
Department of Respiratory Medicine – Intensive Care, Santa Maria Clinic
E-mail: kyllenbrio@yahoo.com
kyllenbrio@ecutel.net
Phone: + +59342893242; + +59387102550; + +59342300532
Adress: Panama y Roca, Panamericana Clinic